ACCELERATORS & EXPERIMENTAL FACILITIES STATUS SUMMARY OF WEEK 28 - 2023

Technical infrastructure – R. Ledru Linac 4 – E. Gousiou PS Booster – G. P. Di Giovanni ISOLDE – *M. Lozano* PS – *M. Fraser* PS – East Area – D. Banerjee PS – nTOF – N. Patronis AD – ELENA – L. Ponce SPS – V. Kain SPS – North Area – D. Banerjee SPS – AWAKE – G. Zevi Della Porta SPS – HiRadMat – Not running – No report Linac 3 – D. Kuechler LEIR – Not running – No report LHC – S. Redaelli & M. Solfaroli CLEAR – W. Farabolini

Technical Infrastructure (TI)										
Facility Coord	inator last weel	Ronan	Ledru							
Facility Coord	inator this weel	Ronan	Ledru							
			Statistic	S						
Alarms	25482									
Phone calls	962	Incoming	543		Outgoing	419				
ODMs	191									
	-	Fa	cility Sta	atus						
Summary	Mon 10/07/23 Evacuation of LHC in stable Fire brigade ad Thu 13/07/23 Loss of 400V Fuse of the 48 CV equipment Thu 13/07/23 LICE TRD co This tank has a contacted to he The final fix w Fri 14/07/23 - 66kV trip at L During the inju differential fau 35 minutes to Sat 15/07/23 - Fire alarm in C Fire Brigade of Intervention fi Sat 15/07/23 - DSS inundation Some water for Neither CV op system Mon 17/07/23 Electrical glito	- 20:04 LHC half- beam, so re- knowledg - 13:22 network E V system I s were aff - 15:51 oling circu negative elp for a so fill need 2 10:50 HC2 durine ection test lt. restore the 00:28 CLEX n site, noth nished at (08:28 n alarm in und in the eration or - 01:00 h on the 2 Quench in	octant 8 nobody i red the a BD1/2U has repla ected bu it trip du pressure olution. days of g works on EHT power hing four 00:54 UX15, "Fosse 2 CV dete 20kV sw the LHO	-1 due to bre n the tunnel. larm due to a 48% ced. t this has bee ne to a leak of stop (next TS on the spare 103, the upp nd beam dump Argon" of U ector cooling	eak glass ala V fault en transpare on a water ta cum piquet I S) e transforme er breaker I to check the X15. has found (Foretaille)	ent for the LHC ank. has been er (EHT103/2E) has opened on e alarm a leak on their) felt at CERN				
Plans										
	<u> </u>	Inter	ention P	equest						
		interv		equest						

Yes / No	Duration	Preferred date/time	
Reason			
Impact			

	Linac 4						
Machine Coor	dinator last	week	E. Gousiou				
Machine Coor	dinator this	week	JB. Lalleme	nt			
			Stat	istics			
Availability	91.7%						
			Facilit	y Status			
Summary	 This week has been the first time Linac4 had issues with a cavity wat pressure sensor. The issue occurred during the night and required HI RF klystron and controls experts and access to the machine. The issue was resolved after 13h30; the main signature of the fault is understood and now the HL-RF teams are following up on ways to prevent, bette diagnose and repair such faults. In the shadow of the issue above two pending interventions took plac BSM1: motor/clutch successfully repaired L4L.RCH.111: FGC replaced – this was part of the investigat of the recurring trips of this corrector; there has been no trip since the FGC replacement. 						
Issues	- [CC	 DTL1, Af ~3 be Th the the the the the re Ac be 	13h30] ter four conse 30min downtin eam was stop the investigation e CCDTL1 car e reading of the erelock). There is a perturbate onnected to the the event occu- tervention fro- cress in the mile hoses, of the placement of ctions for prev- ering followed .111, 17min] wo trips of the uring the CCL ere has been	ecutive trips on the CC me (due to klystron fila oped for investigation b on showed that a faulty avity (used for diagnos he klystron filament cu e was no issue with th ion on the reading cau he same PLC IO. urred during the night a m HL-RF klystron expen- nachine was needed for he needle manometer a the faulty sensor. vention, better diagnos up by the HL-RF team e power converter of th DTL1 intervention, the no trip since.	EDTL1 cavity, each causing ment heating time), the y the HL-RF experts. / water pressure sensor of tics) disturbed electrically irrent (used as an e klystron itself, there was sed by the sensor that is and required the erts and controls experts; or the visual inspection of and for the eventual is and faster repair are is corrector. FGC was replaced and		
Plans	Regular ope	ration		·			
			Interventi	on Request			
Yes / No	Duration	4h		Preferred date/time	Not urgent; to be done in the shadow of another stop		
Reason	Finalise elev	ator re	epair				
Impact	No beam	No beam					

PS Boos	ster	
Machine Coor	dinator last we	k G.P. Di Giovanni
Machine Coor	dinator this w	R. Murillo Garcia
		Beam Scheduled
ISOLDE	Yes	PS Yes
		Beam Availability by Destination (AFT)
ISOLDE	90.6%	PS 90.6%
		Facility Status
Summary	 Not an easy with the P During mo Other than beam cheat 	week for the PSB with most of the downtime caused by the Linac4, and a few issues B itself, see below. of the week, GPS took beam steadily at high current (up to 2 uA). hat, the preparation of the HiRadMat run was completed, and the STAGISOGPS ded to be ready for the upcoming week.
Issues	 On Monda interlockin The p follow Bl2.B3 provid Initially to rect finally tempe On Tuesda When they d situati On Thursda When they d situati On Thursda The lease In the > Bl2. The deb The deb The stop cavi syste At the time CCDTL01 by the BLN The is BT2.N The is 	late afternoon we experienced a few consecutive trips of the BI2.BSW1L1.1 all beams via the BIS: cedure 'PSB injection into less than four rings' (EDMS 2390281) was rigorously d. V1L1.1 interlock masked and injection in R2 inhibited. During this time, the PSB d beam to TOF with R3, although at 15% lower intensity than the nominal beam. PiPO changed an electronic card, but it did not solve the issue although this managed er ring 2 overnight. When the morning after the converter tripped again, the issue was acked down to a loose screw on the temperature sensor which resulted in wrong ture measurements which were causing the equipment to trip. Fixed. 7, the ABT piquet was called to check on a few consecutive trips of the BE2.KFA14L1: not observe any anomaly. We will keep monitoring the h. y we performed the routine access in the PSB to inspect the existing water leak on the inet: k rate remains constant wrt previous weeks (60 ml/min). hadow, a few more interventions were carried out: SW16L4 generator was replaced, as the current was out of specs. b-Train team installed additional diagnostic on the spare chains to continue the gging, following last week's issues. cN-CV team required an intervention on the chilled water towers. This followed a short in the night between Tuesday and Wednesday which caused the trip of the PSB es in sector 5 & 7. EN-CV team cleaned the filters and managed to rebalance the m. The latter was not possible remotely before and it was the main reason to call for ervention. of the restart with beam, and following the long stop due to the issues with the i Linac4, we had unusual high losses in the extraction TL resulting in beam being cut s for GPS only. ue was tracked down to an unexpected charging effect on the recombination kicker "A20 which was pulsing ~2 kV above the setpoint. The fork between the setpoint and uired voltage reduced over time and in ~2 hours it was back to nominal. ABT experts een notified to follow this up. To recover the situation during this period, we adjusted boint to
Plans	Next routin	access for the QFO11 visual check on 27 th July 2023.
		Intervention Request
No	Duration	Preferred date/time
Reason		
Impact		

	ISOLDE						
Machine Supe	ervisor last wee	k	Miguel I	Lozano			
Machine Supe	ervisor this wee	k	Alberto R	odriguez			
			Beam S	cheduled			
GPS	Yes	HRS		No	HIE-ISO	Yes/	
	Bea	am Av	ailability l	by Destination	(AFT)		
GPS	91%	HRS		N/A	HIE-ISO	95.8%	
			Facilit	y Status			
Summary	GPS : Ac bean Smooth HRS: On stand with a lo HIE-ISOLDE :	ns to G run. I by. S wer ga Stable the we from F week.	GHM and L ome inves is flow wer beam to I beek and sta Friday after	A1 Laser Ionize tigations about t e done. SS (22Ne6+ and able beam to Mi rnoon in prepara	d from Monday to the possibility of run d 40Ar11+ at 7.58 l niball (22Ne6+ at 3 ation for the radioad	Sunday night. nning the RFQ MeV/u) during 5.57 MeV/u) trive run this	
Issues	REX 7GAP3 rf	amplif	ier tube re	placed.			
Plans	Target change First radioactiv	on Mo e bear	nday mori n of the 20	ning to be used 23 campaign.	in the Miniball Hg r	un this week.	
			Interventi	on Request			
Yes / No	Duration			Preferred d	ate/time		
Reason							
Impact							

PS							
Machine Coor	dinator last	week M	atthew Fraser				
Machine Coor	dinator this	week Al	exander Husc	hauer			
			Beam Sche	duled			
East Area	Yes	nTOF	Yes	AD	Yes	SPS	Yes
Beam Availability by Destination (AFT)							
AD	83 %	EA N	83%	EA T8	83%	EA T9	83%
nTOF	83%	SPS	81%				
			Facility St	atus			
Summary	 PS-OP te drifting close of the drifting close of the	am forced osed-orbit ng since alerted b smission ie10 p/s u perturbatio parison o source ir bvious iss minute re discussio and see h pus symp n of the p ursday's	to re-steer a to f the PS: 7 th July the an y radiation ala and beam los intil extraction on similar thro f CO taken or the region of sues during vi equest, thanks on with EPC/M now the situati toms of an ele- position of the access and be	unknown arms (PAX: s affected could be r ughout the 14 GeV b the PS fro sual inspec Alexandre ISC expert on evolves ctrical issu SEH23's a cam loss re	t injection and reason 304) on Wed on most cycle e-steered ov cycle: at inje- are cycle in so m 85 – 95 (n ction during T ction during T ction during T ction during t cretin) ts on Friday es before invest before invest ctuation syst cturned to no	d extraction nesday 12 es: n_TOF er night ection and June indic ot conclus Thursday's evening, a stigating fu em was c rmal on E	2 th July F flux limited extraction ates an sive) s access agreed to urther: no arried out AST cycles
Issues	 Water lea dump TD Operation water-flow Larger bu RF loops 	k at 2 cub l48, disco n of TDI47 v warning nch lengt	bic metres per overed during 7 continued as is on TDI47. ths observed o	day on PS Thursday's instructed on 8b4e an	SR-TT2 circuis access and by SY-STI c d fixed: likely	it attribute isolated. lespite con due to ba	d to internal ntinued ad contact in
Plans	Understar	nd and sta	abilise source	of PS clos	ed-orbit drift		
			Intervention	Request			
Yes	Duration	1h	Pro	eferred da	te/time	C: depend lution of th	ing on ne CO drift
Reason	Audio / visua	al inspect	ion of PS to c	neck for is:	sue with MUs	6	
Impact	Access in P	SR (poter	ntially SY) whi	lst PS mai	n units are pu	ulsing	

PS East Area								
Facility Coord	Facility Coordinator last week D. Banerjee							
Facility Coordinator this week J. Bernhard								
				Beam Sch	neduled			
T8	Yes	T9		Yes	T10	Yes	T11	No
	Beam A	Availab	ility l	by Destina	tion (AFT) (General: 90	.6%	
Running T8	28.6%	T 9		28.6%	T10	28.6%	T11	N/A
				Facility S	Status			
Summary	T09: WCTE was checke T10: Good T11: No us	ed and i operation operation operation	tion c re-ste on.	ongoing in I eered on TI	low momentu hursday.	um configura	ation. Beam	on target
Issues								
Plans	• T0 • T1	9: WCT 0: ALIC	E co E ITS	ntinues S3 → BL4S	5			
Intervention Request								
Yes / No	Duration			Р	Preferred dat	te/time		
Reason								
Impact								

PS nTOF								
Facility Coord	inator last w	veek Nikolas F	Nikolas Patronis					
Facility Coord	inator this w	veek Nikolas F	Nikolas Patronis					
Beam Requested								
Yes								
Facility Status								
Summary	Progressing with physics programme according to planning							
Issues	No issues.	No issues.						
Plans	 EAF mea fissi ~30 EAF mea NEA dete 	R1: 30 Si(n,g) means as the second se	surement successfu Tuesday18.07.2023 cm). ²⁴³ Am(n,f) setu urement up to Wed ting. This is 3 weeks on Wednesday 19.0 ew pulses to check	Ily completed. Auxiliary 3. Collimator change (capture to up on Wednesday to stay for nesday morning. 64Ni(n,g) s measurement. 7.2023 to install diamond electronics modifications.				
		Forese	en Beam Stop					
Yes	Duration	5 h	Date/Time	WED 19.07.2023; 09:00-14:00				

AD - ELENA								
Machine Supe	Machine Supervisor last week Laurette Ponce							
Machine Supe	ervisor this wee	k	Pierre F	reyer	muth			
Beam Scheduled								
AD	Yes			ELE	NA	Yes		
	Availability (AFT)							
AD	97.4% ELENA 97.4%				97.4%			
Facility Status								
Summary	 Now two dipol continuous puls Elena H- sour Still looking fo 	es in t sing. ce filar r the c	he DI lin ment bro auses of	e run ke. It AD ii	with forewarnin was replaced t njected intensity	ng 12 pusles instead of hanks to ABP team. y fluctuations.		
Issues	One trip of the	Quads	, it takes	~30r	nin to restart.	-		
Plans	Physics Pbar p	roduct	ion					
		I	nterven	tion F	Request			
Yes / No	Duration				Preferred date	/time		
Reason								
Impact								

			SP	S							
Machine C	oordinator la	ast week	Verena Kain								
Machine C	oordinator th	his week	Arthur Spierer								
			Beam So	Beam Scheduled							
LHC	Yes	NA	Yes	AWAKE	No	HiRadMat	No				
		Beam	Availability b	y Destinatio	on (AFT)						
LHC	~95 %	NA	~83 %	AWAKE	n.a.	HiRadMat	n.a.				
			Facility	Status							
	Relatively go injector comp	Relatively good week for the SPS. The majority of faults were faults of the SPS njector complex with a total availability of about 83 % (Sunday evening).									
	🍸 Faults by Root Cause 🛛 🖉 🖉										
	 Raw Root 	(includes fault Cause (child fa	Faults by Roo is in shadows and chi aults assigned to par	t Cause () Id faults) ent systems, time	in shadow remo	≡ ved)					
	Extractio Machine Interlo	on Systems	1								
	Radio	Frequency	6								
	Power	Converters	3								
	Per	Operation	6								
	Injecto	or Complex				92					
		0	5 10	15 2	0 25	30 35					
	Fault time [h]										
Summary	The main pro LHC filling w scraping was Attempt to pr The TT10 BL quality monit to running in only) coming understood I increase the The TT10 BL meeting on M Several MDs	ogram wa as smooth s typically ut EBC int -M thresh to interloc back afte osses in T threshold -M thresh Monday.	s delivery of p h this week wir 7 %. Also, FT to operation to olds for all bea hanges from the k limits. The 2 er the dedicate TT10. During the s along the er old strategy w	hysics to NA thout any iss physics wa be schedul ams are set he PS lead t h of degrad d MD on Wo his week we tire line to a ill be further	A and physic sues with lo s smooth no ed with NAG to very low o significant ed beam (si ednesday w e therefore a at least 5 m discussed	cs beam to the sses at inject or major issue 52 in week 30 values to act treduction of ingle injection vere due to ne adopted the s Gy in case of in the SPS control	tion. The sto report. as beam availability of or FT on- trategy to issues. bordination				
	of phase jum analysis/com hysteresis co dump mode. Unfortunately flat bottom lo collimator ge the relatively The long par In preparatio intensity was wirescanners the YETS str ion run are a	y the dedi osses was etting stuck complex callel MD of a discusses s need to categy for	d target beam, sure transmiss ion tests and e cated MD with almost entirel k and slow pro- set up require on Thursday w MD however t ed with the BI t be tested with the remaining consideration	test new m ion depende effect of incre- the collima y lost due to gress during d for that M ras postpone he strategy eam. The m the highest scanners. T . The IEFC	ethod for tu ency on veri eased inten tion team to prepeated I g the little re D. ed due the I for the wire nain conclus intensities The potentia will come u	INAC4 long scanners and sity on T4 fo understand HC filling, is maining bea INAC4 long scanners and sion is that the asap to defin al consequen p with an app	naticity e, r beam LHC beam sues with im time with downtime. d LHC high e e or confirm ces for the proved				

	strategy for wire them breaking w Other preparation week and preparation plateau requires (145 A). This was thought to be a from 14 GeV. It assuming 14 Ge checking). Sund was set up.	escanners and ion re with high intensity. ons took also place: aration for oxygen ru s new regulation for as successfully teste limitation for the set turned out however eV dump kicker volt day afternoon the in	un in the near future in t installation of HiRadMa in with 13 GeV injection the main dipoles and al ed. The beam dump sys t up with 13 GeV, as it w that it also arms in time ages (which should be f div cycle with Q26 in pre	he unlikely scenario of at experiment for next . The 13 GeV injection so lower standby current atem SBDS was initially vas supposed to only arm of or 13 GeV, but ine, experts are eparation for the ion run					
Issues	 Freques The edd and the the SPS The SP SFTPR several Not und MCB) TT10 lo 	 Frequently LHC beam splitting from PS not reproducible within one filling The eddy current compensation is not set up for the LHC operational cycle and the H mean drifts significantly at flat bottom. To be discussed during the SPS coordination meeting whether and when to compensate it. The SPS sequence for giving access in point 8 needs to be reviewed. SFTPRO injected 50 buckets off. Ball with the PS – they need to check several other aspects before being able to correct for that. Not understood: RQIF.80400 trips. EPC working on it (suspecting issue with MCB) 							
Plans				-					
		Interventi	on Request						
YES	Duration	4 h	Preferred date/time	n.a.					
Reason	No access nee Montesinos)	ded, but 4 h without	beam to replace Philipp	os amplifier (contact E.					
Impact									
		Interventi	on Request						
YES	Duration	0.5 – 2h	Preferred date/	ttime n.a.					
Reason	CV alarm BA5,	access in the tunne	el required asap.						
Impact									

	SPS North Area									
Facility Co	ord	inator la	ist w	veek	D. Banerjee					
Facility Co	ord	inator th	nis w	veek	J. Bernhard					
					Beam	Scheduled				
H2	Ye	S	H6		Yes	K12	Yes	P42	Yes	
H4	Ye	S	H8		Yes	M2	Yes	TT20	Yes	
		В	eam	Availal	bility by Des	stination (AF	T) Gener	al: 87.5%		
H2	84.	8%	H6		84.8%	K12	84.8%	P42	84.8%	
H4	84.	8%	H8		84.8%	M2	84.8%	TT20	84.8%	
					Faci	lity Status				
Summary Issues		H2/H4/H M2: Mu user. P42/K1: the last request between T2/T4/T change M2: XW electron Area. To P42/K1: be an u investig Some n	H6/H on b 2: Go MD for 1 for 1 n 100 6 Sh d to VCM. ics t v CCM. v CCM. stal ated ated anagn	8: No is eam cho togethe 74 inten 0 and 12 aaring: 3 40 mm 0 061.102 o be rep planned set of tr ole regu et reset	sues, good of ecked for NA ration. Chec r with HSE-F sity from July 20 units. 30 (T2) - 72 (on 19.07. for 2/219/543 nc blaced for who d. ips of P0survitation card the s.	operation we A64mu. Finer RP, no issues y19th on nee (T4) - 45 (T6) <u>NA62 BD ru</u> of working. Be hich access w vey with negl hat gives a w	ek. tuning to ensity for N s up to abo ds to be s with 180 n. eing follow vill be nee igible dow rrong read	be done nex NA62 beam d out 72 units o still evaluated mm T4 targe ved up with B eded with no e vntime, reaso dout from time	t week with the ump run during in T10. Exact , probably it. T4 to be il. Local extraction to North in unknown. Might e to time, being	
Plans		H2: N H4: G H6: A H8: A M2: N	A61/ IF++ TLA: TLA: IA64	SHINE → CM S HGTE S TileCa mu con	continues. S ECAL.), ATLAS MA al \rightarrow MPGD tinues.	ALTA continu TRD, QFIB, 3	ies. Straw Tra	cker RD.		
					Interven	ntion Reques	st			
Yes / No		Duratio	n	No		Preferred da	ate/time	N/A		

SPS AWAKE						
Facility Coordinator last week		Giovanni Zevi Della Porta				
Facility Coordinator this week		-				
Facility Status						
Summary	Bakeout of new Plasma Source. Continue GSM cabling. Laser compressor optics upgrade. Rubidium diagnostics installation					
Issues	Update on CTU timing issue: new firmware installed. Testing in progress.					
Plans	Functional tests to commission new Plasma Source. Pump down laser system and commissioning after optics upgrade.					
Foreseen beam stop						
Yes / No	Duration			date/time		

LHC					
Machine Coor	dinator last we	k S. Redaelli (Mon/Tue/Wed), M. Solfaroli			
Machine Coordinator this week M. Solfaroli (Mon/Tue), J. Wenninger					
	Statistics				
Availability	73.2% Stable Beam Ratio 42.5%			42.5%	
	Facility Status				
Summary	Facility Status Operation for physics with 2464 bunches, intensity ~1.6e11 p/b. Large impact from thunderstorm on Tuesday night, with 2 fills dumped and few heaters fired. Very long fill on Wednesday night/Thursday morning due to unavailability of LINAC4 (RF issue). CMS magnet cooling sensor replaced on thursday afternoon. Magnet re-filling and ramp up took several hours. Several activities during the afternoon and night, waiting for CMS to be back: • ALICE cooling issue fix • Dry run of HB settings • Optics measurements with Xing bumps on IONS cycle • Faster beta* levelling test Trip of 66kV on Friday morning (loss S12, S23, ALICE magnet, IP2 cryo compressor). Once physics restored, on Saturday morning, water leak in ATLAS (access needed) forced to dump prematurely. The investigation took a good part of the day. Stable beams until Sunday 1am when RF trip and quench of few magnets occurred (RQ7/9/10.R4, RQ10.R8, RQX.L8). Following the quench in RQX.L8 the cold compressor was lost and the insulation vacuum pression increased consistently. TDIS temperature always very close to dump threshold during injections, leading to dump if not extracted immediately after injection is completed				
Issues	RQX.L8 insulation vacuum degradation (to be assessed) TDIS temperature interlock 66 kV trip -> loss of ALICE, PCs, cryo compressor (solved) ATLAS water leak (to be followed up) BSRT-B2 sometimes getting blocked BSRA calibration to be adjusted				
Plans	High beta (3/6 km) optics commissioning (Mon) HB background test (Tue) Physics production				
	Intervention Request				
Yes / No	Duration		Preferred date/time		

Linac 3					
Machine Supe	rvisor last week	D. Küchler	D. Küchler		
Machine Supe	rvisor this week	G. Bellodi	G. Bellodi		
Statistics					
Availability	%				
Facility Status					
lon species	Pb				
Summary	 Most of the time good, stable beam from the source (more than 100 eµA out of the RFQ). Some trips of the microwave which have to be followed up for further understanding. Controls for RFQ and tank1 delivered. Successful test of the oxygen inhibit system in the ITL line. 				
Issues					
Plans	Continue to condition the source, after the DSO test (18 July) start to send beam through the linac. Continue the RF setup.				
Intervention Request					
Yes / No	Duration		Preferred date/time		
Reason					
Impact					

CLEAR				
Facility Coordinator last week		Wilfrid Farabolini		
Facility Coordinator this week		Pierre Korysko		
Facility Status				
Summary	Last week was dedicated to Machine Development including: - One-to-One and Dispersion Free Steering Corrections. - Dosimetry studies for Cancer Therapy with VHEE at UHDR. - New beam line preparation work. - Installation of this week experiment.			
Issues	No major issue.			
Plans	This week is dedicated to the following experiment: - Testing a new prototype for an electro-optical near-field monitor to measure the longitudinal bunch profile using electro-optical spectral decoding (EOSD) for FCC-ee (with Karlsruhe Institute of Technology).			